Author's response to reviews

Title: Altered PPP2R2A and Cyclin D1 expression defines a subgroup of aggressive luminal-like breast cancer

Authors:

Francisco Beca (francisco_debeca@dfci.harvard.edu)
Miguel Pereira (miguelpereira@gmail.com)
Jorge F Cameselle-Teijeiro (videoprimaria@mundoro.com)
Diana Martins (dmartins@ibmc.up.pt)
Fernando Schmitt (fschmitt@ipatimup.pt)

Version: 2  Date: 6 October 2014

Author's response to reviews:

Dear editorial team,

As requested, email addresses of all authors have been included on the title page.

Methods section has been clarified with the introduction of the comment: (waiving the need of specify approval for samples from a tumor archive and used in retrospective analysis) after stating the study complies with the Portuguese National Regulative Law for Handling Biological Specimens from tumor banks.

A "conclusion" subtitle for the last paragraph has been included.

Thank you very much.

Best regards,

Francisco Beca